Author’s response to reviews

Title: GLP-1 analogues for neuroprotection after out-of-hospital cardiac arrest: study protocol for a randomized controlled trial

Authors:

Sebastian Wiberg (scwiberg@gmail.com)
Christian Hassager (hassager@dadlnet.dk)
Jakob Hartvig Thomsen (jakob@jht.dk)
Martin Frydland (martin.frydland@me.com)
Dan Hoefsten (dan.eik.hoefsten@regionh.dk)
Thomas Engstroem (thomas.engstroem@regionh.dk)
Lars Koeber (lars.koeber@regionh.dk)
Henrik Schmidt (henrik.schmidt@rsyd.dk)
Jacob Moeller (jem@dadlnet.dk)
Jesper Kjaergaard (jesper.kjaergaard.05@regionh.dk)

Version: 1 Date: 19 May 2016

Author’s response to reviews:

Thank you for your preliminary assessment of our protocol.

As you know the manuscript was submitted in early December 2015, and to the best of our knowledge, the instruction for authors at that time did not mention the SPIRIT check list.

We have waited for the manuscript to be assigned to an editor since December, and have made several enquiries on the progress. We have updated the manuscript according to the SPIRIT guidelines as much as possible, including the SPIRIT checklist, and hope for your understanding that we have a need a swift review since the study will end in the beginning of June.
Please note, that the manuscript have already undergone peer-review by the Danish National Board of Health.

Best Regards,

The authors